Author’s response to reviews

Title: Multimodal therapy in treatment of rectal cancer is associated with improved survival and reduced local recurrence - a retrospective analysis over two decades

Authors:

Armin Wiegering (wiegering_a@ukw.de)
Christoph Isbert (isbert_c@ukw.de)
Ulrich A Dietz (dietz_u@ukw.de)
Volker Kunzmann (kunzmann_v@ukw.de)
Sabine Ackermann (bineackermann@web.de)
Alexander Kerscher (kerscher_a@ukw.de)
Uwe Maeder (maeder_u@ukw.de)
Michael Flentje (flentje_m@ukw.de)
Nicolas Schlegel (schlegel_n@ukw.de)
Joachim Reibetanz (reibetanz_j@ukw.de)
Christoph-Thomas Germer (germer_c@ukw.de)
Ingo Klein (klein_i@ukw.de)

Version: 6 Date: 21 October 2014

Author’s response to reviews:

Review 3

To:
Dr Masahide Ikeguchi
Editor of BMC Cancer

Dear Dr. Ikeguchi, dear Ms Roselyn Remoto,
Thank you for giving us the chance to revise the manuscript with the title

Multimodal therapy in treatment of rectal cancer is associated with improved survival and reduced local recurrence - a retrospective analysis over two decades

We have answered the comments raised by the editor point by point below.

• In your abstract, please rename Purpose as Background, and Introduction as Background in your main manuscript.
  -> Done.

• Please can you clarify whether you received ethics committee approval for your study, please also clarify who granted permission to access data from the registry. Please include all these details in the methods section of your manuscript.
-> We included the statement of the ethic committee and the permission of the access to the data in the methods section.

- Please can you provide individual authors contributions for all 12 authors of this manuscript.
  -> Is now included in the manuscript.

- Acknowledgements
  -> Have been included now.

Best regards
Armin Wiegering